\documentstyle[12pt,amstex,twoside]{article}
\input{macros-course}

\begin{document}
\hlabel{ps-2}{37}               

\pset{-2}{May 15, 1997}{At final}

This summer, I and others will undertake a complete overhaul of 6.042
content and teaching methods.  Your feedback would be of {\bf great
  help} in designing a better course.  Please take 15 minutes to
answer this survey.  Anonymous responses are, of course, fine.le


\begin{problems}

\problem Which of the following courses did you take before or while taking
6.042 (circle)?

Took before: 6.001\ \ 6.033\ \  6.034\ \ 6.041\ \ 6.046\ \ 6.170

Took during: 6.001\ \ 6.033\ \  6.034\ \ 6.041\ \ 6.046\ \ 6.170

  \problem How did you feel about the course content?  Was the total
  amount of material covered too much, just right, or not enough?  Was
  the right proportion of concepts and examples presented?

\vspace{1.6in}

\problem Which (if any) topics were the most fun?  What topics were
deadly boring? 

\vspace{1.6in}

  \problem What material that you learned here do you think will be
  useful in future courses/career?  Which seems pointless?

\vspace*{1.5in}

  \problem Discuss the course readings, textbook and handouts.
  Useful?  Too much detail/not enough?

\vspace{1.5in}

  \problem  What other suggestions due you have about improving the course?


\end{problems}
\end{document}

% Local Variables: 
% mode: latex
% TeX-master: t
% End: 
